<%namespace name="tools" file="tools.tex"/>

\section{Large scale laboratory tests}

\subsection{M1797: Delta Flume 1981}

In 1981, Delta Flume experiments were performed to gain insight in the effect of a dune revetment on the morphological behaviour of the dune. The profile in question is based on a stretch of coast called the Noorderstrand at Schouwen, the Netherlands\citep{Vellinga1981b}. Two large scale experiments (depth scale of 2) were performed, one with and one without dune revetment. The latter is depicted in \autoref{fig:Deltaflume_M1797_T01_final_profiles}.

${tools.image('Deltaflume_M1797/T01/', 'final_profiles.png', 'Final profile of test T01')}

\subsection{M1263 III: Delta Flume 1980-1981}

As a continuation of parts I and II of the M1263 experiment series, large scale tests (depth scales between 1 and 5) were performed in the Delta Flume\citep{Vellinga1984}, with the goal of verifying the relations found at smaller scales. The test result are shown in \autoref{fig:Deltaflume_M1263_III_Test-1_final_profiles} to \autoref{fig:Deltaflume_M1263_III_Test-5_final_profiles}.

${tools.image('Deltaflume_M1263_III/Test-1/', 'final_profiles.png', 'Final profile of test 1')}
${tools.image('Deltaflume_M1263_III/Test-2/', 'final_profiles.png', 'Final profile of test 2')}
${tools.image('Deltaflume_M1263_III/Test-3/', 'final_profiles.png', 'Final profile of test 3')}
${tools.image('Deltaflume_M1263_III/Test-4/', 'final_profiles.png', 'Final profile of test 4')}
${tools.image('Deltaflume_M1263_III/Test-5/', 'final_profiles.png', 'Final profile of test 5')}

\subsection{LIP11D: Delta Flume 1994}

${tools.image('DeltaflumeLIP11D/2E/', 'final_profiles.png', 'Final profile of test 2e')}

\subsection{H4357: Delta Flume 2006}

The 2006 Delta Flume experiments were performed to asses the effect of wave period on dune erosion\citep{VanGent2008}. The tests were performed at a depth scale of 6 and with both Pierson-Moskowitz and double-peaked (DP01 and DP02) wave spectra. Results are shown in \autoref{fig:Deltaflume2006_T01_final_profiles} to \autoref{fig:Deltaflume2006_DP02_final_profiles}.

${tools.image('Deltaflume2006/T01/', 'final_profiles.png', 'Final profile of test T01')}
${tools.image('Deltaflume2006/T02/', 'final_profiles.png', 'Final profile of test T02')}
${tools.image('Deltaflume2006/T03/', 'final_profiles.png', 'Final profile of test T03')}
${tools.image('Deltaflume2006/T04/', 'final_profiles.png', 'Final profile of test T04')}
${tools.image('Deltaflume2006/DP01/', 'final_profiles.png', 'Final profile of test DP01')}
${tools.image('Deltaflume2006/DP02/', 'final_profiles.png', 'Final profile of test DP02')}

${tools.image('Deltaflume2006/DP02/', 'final_profiles.png', 'Final profile of test DP02')}

\subsection{Grosse Wellen Kanal 1986}

${tools.image('GWK86/T01/', 'profiles.png', 'Final profile of test T01')}
${tools.image('GWK86/T02/', 'profiles.png', 'Final profile of test T02')}
${tools.image('GWK86/T03/', 'profiles.png', 'Final profile of test T03')}
${tools.image('GWK86/T04/', 'profiles.png', 'Final profile of test T04')}
${tools.image('GWK86/T05/', 'profiles.png', 'Final profile of test T05')}
${tools.image('GWK86/T06/', 'profiles.png', 'Final profile of test T06')}

\subsection{Grosse Wellen Kanal 1998}

${tools.image('GWK98/A9/', 'final_profiles.png', 'Final profile of test A9')}
${tools.image('GWK98/B2/', 'final_profiles.png', 'Final profile of test B2')}
${tools.image('GWK98/C2/', 'final_profiles.png', 'Final profile of test C2')}
${tools.image('GWK98/F1/', 'final_profiles.png', 'Final profile of test F1')}
${tools.image('GWK98/H2/', 'final_profiles.png', 'Final profile of test H2')}